Author's response to reviews

Title: Psychological, Cultural and Neuroendocrine Profiles of Risk for Preterm Birth

Authors:

R. Jeanne Ruiz (roberta.ruiz@tamhsc.edu)
Alok Kumar Dwivedi (Alok.dwivedi@ttuhsc.edu)
Indika Mallawaarachichi (Indika.mallawaarachichi@ttuhsc.edu)
Hector G Balcazar (hector.g.balcazar@uth.tmc.edu)
Raymond P Stowe (rpstowe@microgenlabs.com)
Kimberly S Ayers (ayers@tamhsc.edu)
Rita Pickler (rita.pickler@cchmc.org)

Version: 4  Date: 12 August 2015

Author's response to reviews: see over
August 12, 2015

Mr. Ian Dominique Trinidad on behalf of
Dr Alexander Heazell
Editorial at Biomedcentral
BMC Pregnancy and Childbirth

Dear Editors,

I am submitting the paper with revisions as directed on the email of August 7th, 2015. The following points have been changed:

- I have removed the authors’ titles from the title page.
- I have changed the Implications section to reflect conclusions.
- The figure legends are now in the main manuscript after the references and are removed from the figure files.
- In the methods sections, there has been a sentence that describes the method for the blood collection, specifically in an EDTA tube.

I look forward to next steps to get the manuscript published.

Thank you,

Dr R. Jeanne Ruiz, PhD